

Delft University of Technology

Drivers of Interannual Sea Level Variability on the Northwestern European Shelf

Hermans, Tim H. J.; Le Bars, Dewi; Katsman, Caroline A.; Camargo, Carolina M. L.; Gerkema, Theo; Calafat, Francisco M.; Tinker, Jonathan; Slangen, Aimée B. A.

DOI 10.1029/2020JC016325

Publication date 2020

Document Version Final published version

Published in Journal of Geophysical Research: Oceans

Citation (APA) Hermans, T. H. J., Le Bars, D., Katsman, C. A., Camargo, C. M. L., Gerkema, T., Calafat, F. M., Tinker, J., & Slangen, A. B. A. (2020). Drivers of Interannual Sea Level Variability on the Northwestern European https://doi.org/10.1029/2020JC016325

Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.

JGR Oceans



RESEARCH ARTICLE

10.1029/2020JC016325

Key Points:

- We use a ROMS setup to quantify the contributions of different drivers to interannual sea level variability on the NWES for 1995–2018
- Atmospheric variability is the main driver, whereas ocean variability drives small and uniform sea level variability on the shelf
- Wind drives large variability in the North Sea, north of the United Kingdom and around Norway, while the inverse barometer effect dominates elsewhere

Supporting Information:

Supporting Information S1

Correspondence to:

T. H. J. Hermans, tim.hermans@nioz.nl

Citation:

Hermans, T. H. J., Le Bars, D., Katsman, C. A., Camargo, C. M. L., Gerkema, T., Calafat, F. M., et al. (2020). Drivers of interannual sea level variability on the northwestern European shelf. *Journal of Geophysical Research: Oceans, 125*, e2020JC016325. https://doi.org/10.1029/2020JC016325

Received 15 APR 2020 Accepted 21 SEP 2020 Accepted article online 22 SEP 2020

©2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Drivers of Interannual Sea Level Variability on the Northwestern European Shelf

Tim H. J. Hermans^{1,2}, Dewi Le Bars³, Caroline A. Katsman⁴, Carolina M. L. Camargo^{1,2}, Theo Gerkema¹, Francisco M. Calafat⁵, Jonathan Tinker⁶, and Aimée B. A. Slangen¹

¹NIOZ Royal Netherlands Institute for Sea Research, Department of Estuarine and Delta Systems, Utrecht University, Yerseke, The Netherlands, ²Department of Geosciences and Remote Sensing, Delft University of Technology, Delft, The Netherlands, ³Royal Netherlands Meteorological Institute (KNMI), De Bilt, The Netherlands, ⁴Department of Hydraulic Engineering, Delft University of Technology, Delft, The Netherlands, ⁵Department of Marine Physics and Ocean Climate, National Oceanography Centre, Liverpool, UK, ⁶Met Office Hadley Centre, Exeter, UK

Abstract Sea level on the northwestern European shelf (NWES) varies substantially from year to year. Removing explained parts of interannual sea level variability from observations helps to improve estimates of long-term sea level trends. To this end, the contributions of different drivers to interannual sea level variability need to be understood and quantified. We quantified these contributions for the entire NWES by performing sensitivity experiments with a high-resolution configuration of the Regional Ocean Modeling System (ROMS). The lateral and atmospheric boundary conditions were derived from reanalyses. We compared our model results with satellite altimetry data and used our sensitivity experiments to show that nonlinear feedbacks cause only minor interannual sea level variability on the shelf. This indicates that our experiments can be used to separate the effects of different drivers. We find that wind dominates the variability of annual mean sea level in the southern and eastern North Sea (up to 4.7-cm standard deviation), whereas the inverse barometer effect dominates elsewhere on the NWES (up to 1.7-cm standard deviation). In contrast, forcing at the lateral ocean boundaries results in small and coherent variability on the shelf (0.5-cm standard deviation). Variability driven by buoyancy fluxes ranges from 0.5- to 1.3-cm standard deviation. The results of our sensitivity experiments explain the (anti)correlation between interannual sea level variability at different locations on the NWES and can be used to estimate sea level rise from observations in this region with higher accuracy.

Plain Language Summary Sea level on the continental shelf northwest of Europe is rising in the long term but is also varying strongly from year to year. This makes it difficult to determine the rate of sea level rise from observations. To improve long-term trends computed from sea level observations, the causes of short-term sea level variability need to be understood. Therefore, we test the influences of different components of the atmosphere and ocean on year-to-year sea level variability, using a numerical ocean model for northwestern Europe. We find that the varying strength and direction of winds causes large variability of sea level in the southern and eastern North Sea. In other places on the continental shelf, sea level variability is mainly influenced by the variability of atmospheric pressure. We also find that the ocean outside our model domain drives small and uniform sea level variability on the continental shelf and that there is a moderate influence of solar radiation, precipitation, and evaporation. Our results help to understand how sea level varies at different locations in northwestern Europe and to obtain better estimates of the rate of long-term sea level rise.

1. Introduction

Sea level on the northwestern European shelf (NWES) has been rising since the nineteenth century (Wahl et al., 2013), and this rise is projected to accelerate in the coming centuries (e.g., Church et al., 2013; Oppenheimer et al., 2019). Fluctuations of sea level on various time scales complicate detecting sea level trends from observations, and sea level variability will continue to be the dominant source of uncertainty of sea level change for the coming decades (Palmer et al., 2018). Removing explained parts of variability from sea level records can lead to more accurate estimates of trends and accelerations (e.g., Calafat & Chambers, 2013; Dangendorf et al., 2014; Gerkema & Duran-Matute, 2017; Hogarth et al., 2020;



Thompson, 1986; Wahl et al., 2013). Therefore, the processes driving interannual to multidecadal sea level variability on the NWES need to be understood.

Drivers of sea level variability on the NWES have mainly been studied using the long tide gauge (TG) records that are available in the region. Wahl et al. (2013) found that interannual sea level variability is largest at TGs in the southeastern North Sea. Using linear regression, Dangendorf et al. (2014) showed that a large part of the interannual sea level variability at TGs in the southeastern North Sea can be explained by variability of wind stress, and at the western U.K. and Norwegian coastlines by the inverse barometer (IB) effect (Stammer & Huttemann, 2008). Additionally, a link has been found between the North Atlantic Oscillation (NAO) and sea level on the NWES (Chen et al., 2014; Tsimplis & Shaw, 2008; Tsimplis et al., 2005; Wakelin et al., 2003; Woolf et al., 2003).

On annual to multidecadal time scales, sea level at several locations on the NWES is significantly correlated with sea level along the eastern boundary of the North Atlantic. Miller and Douglas (2007) and Woodworth et al. (2010) correlated sea level at Brest (France) and Cascais (Portugal) to sea level pressure (SLP) over the subtropical North Atlantic and suggested a link between multidecadal sea level variability and mass redistribution related to gyre-scale adjustments. Others showed that sea level variability is coherent from northwestern Africa to the NWES and strongly correlated with integrated longshore winds (LSWs) on time scales of 1 year to decades (Calafat et al., 2012; Sturges & Douglas, 2011). This suggests an important role of boundary dynamics and, in particular, wind-driven coastally trapped waves (CTWs). Dangendorf et al. (2014) hypothesized that CTWs also affect sea level variability in the North Sea. Using the correlation between sea level in the North Sea and the eastern North Atlantic, Frederikse, Riva, Slobbe, et al. (2016) closed the sea level budget for the North Sea for 1958–2014. More recently, Chafik et al. (2019) argued that steric anomalies in the North Atlantic are communicated to the NWES through Ekman transport and affect inter-annual to decadal sea level variability, since sea level, along-slope winds, and subpolar gyre properties covaried during 1993–2016.

These studies have explained sea level variability on the NWES at interannual to multidecadal time scales using statistical relationships between observed sea level at different locations and between sea level and a variety of oceanic and atmospheric drivers. Many of these drivers, such as SLP, winds, gyre circulation, precipitation, and the NAO, are correlated themselves (Woodworth et al., 2009). This makes the correlations between observed sea level and sea level drivers challenging to interpret. Additionally, since TGs are sparsely distributed and located at the coast, TGs alone do not allow us to characterize sea level variability spatially and determine whether observed signals are purely coastal or originate in the open ocean. Satellite altimetry has a better spatial coverage than TGs but can be less accurate in coastal zones due to altimetric corrections and land contamination.

Instead of using observations, Roberts et al. (2016) ran sensitivity experiments with a global ocean model to link interannual sea level variability to different modes of climate variability. More recently, Tinker et al. (2020) used a high-resolution regional ocean model to explore the influence of atmospheric forcing and lateral ocean boundary conditions on interannual sea level variability on the NWES. However, they used a climate control simulation and did not investigate the effect of specific drivers. Hence, the contributions of different drivers to interannual sea level variability in the satellite altimetry era are not yet fully clear for the entire NWES region.

Here, we investigate different drivers of interannual sea level variability for a period in the satellite altimetry era (1995–2018). To this end, we developed a high-resolution setup of the Regional Ocean Modeling System (ROMS; Shchepetkin & McWillams, 2005) for the NWES. We derived boundary conditions for our model from ocean and atmosphere reanalyses (GLORYS12v1 and ERA5, respectively). By separately varying the lateral ocean boundaries, SLP, winds, and buoyancy fluxes in our model, we quantify the effects of these drivers on interannual sea level variability.

We explain our ROMS setup and methods in section 2 and show that our model simulations agree well with satellite altimetry data in section 3. We show the influence of forcing at the lateral ocean boundaries, SLP, winds, and buoyancy fluxes on interannual sea level variability and discuss correlations between variability at different coastal locations in section 4. We end with a discussion in section 5 and our conclusions in section 6.





Figure 1. Map of the ROMS domain showing the smoothed ETOPO1 bathymetry (m) and the land mask (gray) of the model setup. The red contours indicate the 80- and 200-m isobaths. Several key regions are indicated.

2. Methods and Data

Here we introduce our ROMS setup for the NWES (section 2.1) and the reanalysis data that we prescribed at the boundaries of ROMS (section 2.2). Additionally, we discuss the satellite altimetry data that we used to evaluate our model (section 2.3) and present the sensitivity experiments that were used to study the influences of different drivers of interannual sea level variability (section 2.4).

2.1. ROMS Setup for the NWES

ROMS is a primitive-equation, curvilinear, hydrostatic, Boussinesq, and free-surface ROMS (Shchepetkin & McWillams, 2005). It runs on a staggered Arakawa C grid and has stretched terrain-following vertical coordinates (Song & Haidvogel, 1994). Terrain-following coordinates are particularly suited for modeling regions with large bottom topography gradients such as the slope from the deep ocean toward the NWES.

Our model domain extends from 36° N to 62° N and from 20° W to 10° E and has four lateral open boundaries (Figure 1). Our model has 30 vertical levels and a horizontal resolution of $(1/8)^{\circ}$ by $(1/8)^{\circ}$ (~11 km), so resolves eddies in the open ocean. With this resolution, the English Channel and

Irish Sea are at least four to five grid points wide. The bathymetry (Figure 1), derived from ETOPO1 (Amante & Eakins, 2009), was smoothed with grid stiffness $rx_0 = 0.13$, which reduces horizontal pressure gradient errors (Beckmann & Haidvogel, 1993). We used mixing along geopotential surfaces.

2.2. ROMS Boundary Conditions From Reanalysis Data

We derived surface and lateral boundary conditions for ROMS from reanalysis data for 1993–2018. For the lateral ocean boundaries, we obtained monthly mean values of sea surface height (SSH), zonal and meridional currents, and temperature and salinity from GLORYS12v1 (Fernandez & Lellouche, 2018). GLORYS12v1 is a global ocean reanalysis with a horizontal resolution of (1/12)° by (1/12)° (~8 km) and covers 1993–2018. Its model component NEMO (Madec & NEMO Team, 2016) is driven by atmospheric forcing from the ERA-Interim reanalysis (Dee et al., 2011). GLORYS12v1 assimilates satellite sea surface temperature, in situ temperature and salinity profiles, and along-track satellite altimetry data corrected for the IB effect and the 18.61-year lunar nodal cycle (Fernandez & Lellouche, 2018).

We imposed Chapman conditions for the free surface (Chapman, 1985), Flather radiation conditions for barotropic velocities (Flather, 1976), and mixed-radiation nudging for baroclinic velocities and ocean temperature and salinity (Marchesiello et al., 2001; Orlanski, 1976). We nudged temperature and salinity toward outgoing nudging at a time scale of 1 year and ingoing nudging at a time scale of 1 day, using a 20-point relaxation zone at the lateral boundaries of the domain. Baroclinic velocities were nudged at the boundaries only. To suppress computational noise, we also applied a 20-point sponge layer, multiplying the diffusion and viscosity by a factor that linearly increases to 10 at the lateral boundaries.

ROMS and NEMO (GLORYS12v1) are both Boussinesq ocean models, meaning that local steric effects are modeled prognostically but non-Boussinesq steric effects are not (Griffies & Greatbatch, 2012). Although non-Boussinesq steric effects have a negligible imprint on monthly or longer regional sea level patterns, they do impact global mean sea level (Greatbatch, 1994; Griffies & Greatbatch, 2012). The global mean sea level in GLORYS12v1 was corrected by adding the global mean steric effect to the SSH diagnostically (Lellouche et al., 2018). This signal may enter our model via the lateral boundary conditions but has only small interannual variability.

The atmospheric boundary conditions were derived from ERA5 (Copernicus Climate Change Service (C3S), 2017). ERA5 is based on the IFS41r2 Earth system model and is available for 1979 to a few months before present. The atmospheric forcing was applied using bulk formulae based on (Fairall et al., 1996). We prescribed solar radiation daily using an idealized diurnal cycle and precipitation 12-hourly. Winds at 10 m above sea level, atmospheric pressure, cloud coverage, air temperature, and relative air humidity at 2 m above sea level were prescribed on a 6-hourly basis.



Table 1	
Overview of the ROMS Sensitivity Experiments	

Experiment	Interannually varying	Repeated annual cycle	Discussed in
Exp_AllVar (reference)	All boundaries	None	Sections 4.1-4.4
Exp_AtmosVar	Surface boundaries	Lateral boundaries	Sections 4.1 and 4.2
Exp_OceanVar	Lateral boundaries	Surface boundaries	Sections 4.1, 4.3, and 4.4
Exp_NoneVar	None	Lateral boundaries and surface boundaries	Section 4.1
Exp_SLPVar	SLP	Winds, buoyancy fluxes and lateral boundaries	Sections 4.2–4.4
Exp_WindVar	Winds	SLP, buoyancy fluxes and lateral boundaries	Sections 4.2-4.4
Exp_BuoyVar	Buoyancy fluxes: solar radiation, precipitation, cloud	SLP, winds, and lateral boundaries	Sections 4.2-4.4
	coverage, air temperature, and relative humidity		

Note. We used a climatological annual cycle for river run-off in every experiment.

We also applied a climatological river run-off based on the river discharge data set of Dai (2017). This means that our model does not capture interannual variability due to freshwater inflow from rivers. We also omit tides in our model. River run-off and tides may contribute to interannual sea level variability but require a model setup with a higher horizontal resolution to be fully resolved.

2.3. Satellite Altimetry Data Used for Model Evaluation

We evaluated model results against sea level observations from the Level-4 gridded Ssalto/Duacs multimission satellite product ($[1/4]^{\circ}$ by $[1/4]^{\circ}$) of AVISO (AVISO, 2019). This product is available for 1993 to a few months before present. It has been corrected for the IB effect and for the nodal cycle through standard tidal corrections. Furthermore, we assume that barystatic effects on interannual sea level variability are small compared to the effects of regional ocean dynamics, allowing for direct comparison with ROMS.

In addition, we evaluated our simulation of mean dynamic topography (MDT), which reflects the average strength of geostrophic currents. We used the CNES MDT CLS18 product (CNES, 2019), which is an estimate of the mean SSH above the geoid, and is given at a $(1/8)^{\circ}$ by $(1/8)^{\circ}$ resolution. It is based on altimetry data, in situ measurements of temperature and salinity, and the GOCO5S geoid model and covers the period 1993–2012.

Although the satellite altimetry data is given on $[1/4]^{\circ}$ and $[1/8]^{\circ}$ grids, respectively, its effective resolution is lower than our model resolution. Hence, small-scale features in the model results are difficult to evaluate. Additionally, land contamination leads to a lower accuracy of satellite altimetry in coastal regions (Andersen & Scharroo, 2011). Finally, satellite altimetry products are usually corrected for the IB effect, tides, and high-frequency winds (e.g., Ponte & Ray, 2002), leading to reduced accuracy in regions on the NWES where these corrections are difficult to model. Therefore, in section 3 we also compare our model results with GLORYS12v1.

2.4. ROMS Sensitivity Experiments

We decomposed sea level variability using six sensitivity experiments with different lateral and surface boundary conditions (Table 1). For the reference experiment, we varied all lateral and surface boundary conditions (*Exp_AllVar*). For the sensitivity experiments, we prescribed ROMS with an interannually varying annual cycle for the driver under investigation and with a repeated mean annual cycle for the others.

We decomposed total interannual sea level variability (*Exp_AllVar*) into variability due to the lateral ocean boundary conditions, atmospheric forcing, and intrinsic variability (*Exp_OceanVar*, *Exp_AtmosVar*, and *Exp_NoneVar*). The influence of atmospheric forcing was further decomposed into the influences of SLP, winds, and buoyancy fluxes (*Exp_SLPVar*, *Exp_WindVar*, and *Exp_BuoyVar*). The SSH from GLORYS12v1 does not include the IB effect. Hence, when we include SLP in ROMS, the lateral boundary conditions from GLORYS12v1 are compensated using the PRESS_COMPENSATE option in ROMS.

All runs were initialized with the final year of *Exp_NoneVar*, that is, the sensitivity experiment in which both the atmospheric forcing and the lateral ocean boundary conditions are a repeated mean annual cycle. We calculated the mean annual cycle of the reanalysis data from the years 1993–1998, instead of the full period 1993–2018. This was done to minimize spin-up effects toward a changed climate state, as the mean state of



1993–1998 is more compatible with the boundary conditions in 1993. Additionally, we discarded the first two model years (1993 and 1994) to avoid any remaining spin-up effects. Therefore, in this manuscript we show model results for 1995–2018 (24 years). Spun-up sensitivity experiments show that results depend marginally on the period used to derive the climatological forcing.

Some of the bulk parameterizations of atmospheric forcing (section 2.2) depend on variables that vary in all sensitivity experiments. Thus, buoyancy fluxes may still vary interannually in sensitivity experiments other than *Exp_BuoyVar*. For example, the variability of SLP (*Exp_SLPVar*) affects air density and specific humidity, which in turn affect buoyancy fluxes. Since these feedbacks are ultimately caused by the interannual variability of the driver under investigation, we do not consider them part of *Exp_BuoyVar*.

Additionally, the response of the ocean to the various components of forcing at the lateral and surface boundaries depends on its state (temperature, salinity, and circulation). For example, the heat taken up by the ocean varies interannually in *Exp_BuoyVar* but is redistributed by the same wind-driven circulation each year. Hence, due to nonlinear feedbacks, the interannual sea level variability in *Exp_BuoyVar* may differ from the actual buoyancy-driven variability in *Exp_AllVar*. This applies to the other sensitivity experiments as well. However, we show in section 4 that such nonlinear feedbacks result in only small interannual sea level variability on the shelf.

3. Evaluation of Model Results

In this section, we evaluate simulated MDT against CNES MDT CLS18 (both as anomalies with respect to their area-weighted regional mean) and interannual sea level variability against the AVISO product. Additionally, we compare our simulations with the GLORYS12v1 reanalysis. We evaluate a variant of *Exp_AllVar* that excludes the effect of SLP. We compared the different fields quantitatively using pattern correlation coefficients (PCCs—the spatial equivalent of Pearson's correlation coefficient) and root-mean-square errors (RMSEs). We used a hypothesis test to determine statistical significance (p < 0.05, with *p* calculated using a two-sided *t* distribution).

ROMS MDT matches well with GLORYS12v1 MDT and CNES MDT CLS18 (Figures 2a, 2c, and 2e). It has PCCs of 0.95 and 0.93 and RMSEs of 3.76 and 4.34 cm with the other products, respectively. All three fields have a north to south gradient of low to high MDT anomalies. The MDT is higher than the regional mean on most of the shelf and largest in the southeastern North Sea. All fields show lower MDT in the Norwegian Trench and the northern North Sea. In ROMS and GLORYS12v1, the 0-cm MDT contour realistically follows the northern coast of Scotland and approximately traces the 80-m isobath into the North Sea.

Since GLORYS12v1 assimilates along-track satellite altimetry, its MDT is similar to the satellite altimetry data. However, in some areas, ROMS and GLORYS12v1 MDT match well but differ from CNES MDT CLS18. For example, ROMS and GLORYS12v1 differ only 1 cm near the coast of Scotland and Denmark, whereas CNES MDT CLS18 is up to 6 cm lower. This could be due to limitations of the satellite altimetry data (see section 2.3).

ROMS MDT differs from the other products near the Rockall Bank and in the southwest of the domain, so here geostrophic currents may not be well reproduced. Our model also has stronger MDT gradients across the shelf break, particularly at the Celtic and Armorican slopes. Such gradients indicate along-slope geostrophic currents, which have indeed been identified in this region with moorings and numerical models (Friocourt et al., 2008; Pingree & Le Cann, 1989) and may be affected by bathymetry smoothing. CNES MDT CLS18 shows no gradient in this area, but its resolution is likely insufficient to assess such a feature. Despite its data assimilation, GLORYS12v1 has a somewhat lower MDT along this part of the shelf break. Other high-resolution regional ocean models with different boundary conditions also show, to various extents, reduced MDT along the Celtic and Armorican slopes (e.g., Hermans et al., 2020; Tinker et al., 2020).

Next, we compare the magnitude of interannual sea level variability in our model with the observations. To this end, we calculated the standard deviation (SD) of linearly detrended, annual mean sea level. *Exp_AllVar* reproduces the observed SDs well (Figures 2b, 2d, and 2f). It has PCCs of 0.80 and 0.78 and RMSEs of 0.49 and 0.53 cm with respect to GLORYS12v1 and AVISO. All three fields have large interannual sea level variability in deep waters, such as the Iceland Basin, the Rockall Trough, and the Bay of Biscay. However, the variability in ROMS (Figure 2b) is larger than in GLORYS12v1 and AVISO (Figures 2d and 2f) northwest





Figure 2. Mean dynamic topography anomalies with respect to the area-weighted regional mean (cm) (left panels) and interannual sea level variability as the SD of detrended, annual mean sea level (cm) (right panels) for (a and b) ROMS, (c and d) GLORYS12v1, and (e and f) CNES MDT CLS18 and AVISO. All products have been interpolated to the (1/8)° grid of ROMS. The PCCs and RMSEs of ROMS with respect to GLORYS12v1 and the satellite altimetry

products are indicated. The black contours indicate 0-cm MDT, and the red contours indicate the 80- and 200-m isobaths of ROMS (a and b) and GLORYS12v1 (c-f).

of Portugal and generally smaller in the rest of the open ocean. Sea level variability is small near the shelf break and the Norwegian Trench, with all three fields showing a similar magnitude of 0.7- to 1.0-cm SD. This is encouraging, since the shelf break might play an important role for (decadal) sea level variability on the shelf (Dangendorf et al., 2014).

The spatial patterns of interannual sea level variability also agree well on the shelf. As in GLORYS12v1 and AVISO, the largest variability in ROMS is found in the German Bight. The variability in the German Bight is somewhat higher in ROMS than in GLORYS12v1 and AVISO (up to 4.9-, 4.3-, and 4-cm SDs, respectively). In the central North Sea, ROMS is closer to AVISO, whereas AVISO has somewhat larger sea level variability in





Figure 3. Pointwise correlation coefficients (–) between detrended, annual mean sea level (1995–2018) in (a) ROMS and AVISO and (b) GLORYS12v1 and AVISO, and (c) detrended sea level with respect to the time mean (cm) at sample locations for ROMS (red, solid) and AVISO (black, dashed), with their correlation coefficients. The black stipples in (a) and (b) indicate statistically insignificant correlation coefficients ($p \ge 0.05$); isobaths as in Figure 2.

the northern North Sea and around the United Kingdom than both ROMS and GLORYS12v1. Additionally, unlike ROMS and GLORYS12v1, AVISO shows a few patches of high variability (up to ~3-cm SD) in the English Channel. These differences could be the result of inaccurate tidal corrections or processes that are not represented in ROMS and GLORYS12v1.

The high pointwise correlation between ROMS and AVISO (Figure 3a) corroborates the agreement seen in Figure 2. Correlation coefficients are high and statistically significant in most of the domain. The insignificant correlations in deep parts of the open ocean (Figure 1) are likely because of the intrinsic variability in these regions (see section 4.1 and Figure 5b). Intrinsic variability in ROMS does not have exactly the same spatial pattern as in the observations and is not expected to have the same phase. In the English Channel, the Irish Sea, and along parts of the shelf break, the correlations between ROMS and AVISO are lower than on the rest of the shelf, which could be due to limitations of the satellite altimetry data (see section 2.3). This is further supported by the pointwise correlation between GLORYS12v1 and AVISO, which is also reduced around the United Kingdom and parts of the shelf break (Figure 3b).

To further demonstrate the high correlations between sea level in ROMS and AVISO, we show the time series for a few sample TG locations at the coast (Figure 3c). Correlation coefficients range from 0.64 to 0.84 for these coastal locations, despite land contamination of the altimetry. The time series in Figure 3c also show how much sea level varies. For example, interannual variations are up to 11.14 cm (Cuxhaven), while longterm trends are only in the order of mm/year (Wahl et al., 2013).





Figure 4. Correlation coefficients (–) between detrended annual mean sea level (1995–2018) at Brest and at all other grid cells for (a) ROMS and (b) AVISO, and cross correlations for AVISO along (c) a meridional transect in the open ocean (25°W) and (d) along the 200-m isobath. Distances (km) along the transect and isobath are indicated. Stipples and isobaths as in Figure 3.

Finally, we compare the correlations of detrended sea level at Brest with detrended sea level at all other grid cells in ROMS and AVISO (Figures 4a and 4b), since the spatial correlation pattern in the eastern North Atlantic and on the NWES has been discussed in a number of previous studies (Calafat et al., 2012, 2013; Chafik et al., 2019; Dangendorf et al., 2014; Frederikse, Riva, Slobbe, et al., 2016; Hogarth et al., 2020; Sturges & Douglas, 2011). The resulting patterns of ROMS and AVISO agree well within the model domain. High correlations (>0.6) with sea level at Brest extend from northwestern Africa to Norway, with the highest correlations found primarily along the continental slope. In both ROMS and AVISO, sea level at Brest becomes progressively less correlated with sea level toward the southern and eastern North Sea. It is negatively correlated with sea level in the open ocean in the west and northwest of the domain, although this anticorrelation is mostly statistically insignificant. To highlight the unique character of the spatial coherence of coastal sea level variability, we compare the cross correlations of observed sea level along the continental slope and along a meridional transect in the open ocean ($25^{\circ}W$) (Figures 4c and 4d). This shows that the correlation scales along the slope are several times longer than those in the open ocean, suggesting shelf-sea dynamics play an important role in driving spatial coherence on the NWES.

We conclude that, overall, interannual sea level variability in ROMS, GLORYS12v1, and AVISO is in good agreement (Figures 2–4). This indicates we can use our ROMS setup to investigate the different drivers of interannual sea level variability in the NWES region (section 4).





Figure 5. Sea level variability in different ROMS experiments (Table 1) as the SD (cm) of detrended annual means (1995–2018) for (a) *Exp_AllVar*, (b) *Exp_NoneVar*, (c) *Exp_AtmosVar*, (d) *Exp_OceanVar*, (e) the linear sum of time series in *Exp_NoneVar*, *Exp_AtmosVar*, and *Exp_OceanVar*, and (f) the residual SD of *Exp_AllVar* minus the linear sum of components. Isobaths as in Figure 1.

4. The Contributions of Different Drivers to Interannual Sea Level Variability

Using the sensitivity experiments in Table 1, we first isolate the contributions of intrinsic sea level variability, atmospheric forcing, and the lateral ocean boundaries (section 4.1). Then, we split up the contribution of atmospheric forcing into the effects of SLP, winds, and buoyancy fluxes (section 4.2). Finally, we show the relative contribution of different drivers to sea level at a few coastal locations (section 4.3) and relate these to the correlation between sea level at different locations (section 4.4).

4.1. Intrinsic Variability, Atmospheric Forcing, and Lateral Ocean Boundaries

In contrast to section 3, here we included the effect of SLP in *Exp_AllVar* (Figure 5a). The spatial patterns of sea level variability in Figures 2b and 5a are similar, but not exactly the same due to covariance. For example, including SLP leads to larger sea level variability in most locations but to slightly lower sea level variability in the central North Sea.

The interannual sea level variability in *Exp_NoneVar* (Figure 5b) has SDs of up to 2.7 cm in the deep ocean (Rockall Trough) but is negligible (<0.1-cm SD) on the shelf. Since all forcing in *Exp_NoneVar* is a repeated mean annual cycle, the resulting variability must be due to spontaneously generated nonlinear processes, such as mesoscale eddies. Therefore, we refer to this variability as intrinsic. A model with tides and

sufficient resolution to resolve shelf-sea eddies may simulate larger intrinsic sea level variability on the shelf. The open ocean variability in *Exp_NoneVar* is lower than the intrinsic variability found in previous studies (e.g., Sérazin et al., 2015) and is negligible at the lateral boundaries. This is the result of the lateral boundary conditions, which were prescribed as a repeated annual cycle in *Exp_NoneVar*. The other sensitivity experiments also have intrinsic variability, so the sea level variability due to forcing at the lateral and surface boundaries is easier to identify on the shelf than in the open ocean.

 $Exp_AtmosVar$ (Figure 5c), in which only the atmospheric forcing varies interannually, shows that the atmospheric forcing is the dominant driver of sea level variability on the shelf. The resulting spatial pattern on the shelf compares well to Exp_AllVar (Figures 2b and 5a) and the observations (Figure 2f). Sea level variability in $Exp_AtmosVar$ is highest in the German Bight (~5.6-cm SD) and is also relatively high around Scotland and Norway (~2.4-cm SD). Intrinsic sea level variability is present in the open ocean and is enhanced by the local effects of atmospheric variability. At the lateral boundaries, the interannual sea level variability is nudged toward the variability due to SLP, since a repeated annual cycle from GLORYS12v1 was used combined with the PRESS_COMPENSATE option (explained in section 2.4).

 $Exp_OceanVar$ (Figure 5d), in which only the lateral boundary conditions vary interannually, has only minor interannual sea level variability of around 0.5-cm SD on the shelf. The variability is mostly uniform on the shelf, as demonstrated by the high correlations between detrended sea level at Brest and elsewhere shown in supporting information Figure S1. By interannually varying the boundary conditions at each lateral boundary separately, we find that the sea level variability on the shelf is caused mostly by the variation of the western and southern boundaries of our model domain (see supporting information Figure S2). Variations in subtropical gyre strength may affect transport through the western and southern boundaries, but we did not find a significant correlation between sea level in $Exp_OceanVar$ and the North Atlantic barotropic stream function in GLORYS12v1. In the open ocean, interannual sea level variability due to the varying lateral boundary conditions is much larger than on the shelf. Geostrophic currents along the continental slope can hinder the propagation of open ocean signals to the coast (Huthnance, 1981). In addition to the intrinsic sea level variability in the open ocean, the lateral ocean boundaries drive strong sea level variability along the western boundary, in the Iceland Basin (~4-cm SD) and east of the Faroe Islands (~2.6-cm SD).

Next, we calculated SDs for the linear sum of the detrended sea level time series in *Exp_NoneVar*, *Exp_AtmosVar*, and *Exp_OceanVar* (Figure 5e). We compared this to the variability in *Exp_AllVar* (Figure 5a), in which all forcing varies interannually simultaneously. The SDs for *Exp_AllVar* and the linear sum are similar on the shelf (Figures 5a and 5e), with differences well below 0.2 cm (Figure 5f). The linear sum also has the same phase, as indicated by the high pointwise correlation with *Exp_AllVar* (see supporting information Figure S3). This indicates that to first order, sea level variability due to the atmospheric and oceanic drivers combines linearly on the shelf. This also holds for the forcing at the separate lateral ocean boundaries (supporting information Figures S2 and S3). Since all sensitivity experiments include intrinsic sea level variability, differences between *Exp_AllVar* and the linear sum are larger in the open ocean (Figure 5f).

4.2. SLP, Winds, and Buoyancy Fluxes

We further decomposed the interannual sea level variability in *Exp_AtmosVar* (Figure 5c, repeated in Figure 6a) into variability induced by SLP (*Exp_SLPVar*), winds (*Exp_Windvar*), and buoyancy fluxes (*Exp_BuoyVar*).

 Exp_SLPVar (Figure 6b) shows the effect of SLP on interannual sea level variability. Except for the intrinsic variability in the open ocean, the resulting variability agrees with the variability of SLP in ERA5 (~1 cm sea level for 100-Pa SLP, see supporting information Figure S4). Thus, interannual sea level variability in Exp_SLPVar is mainly due to the IB effect. The resulting sea level variability is smallest in the south of the domain and in the southeastern North Sea (~1-cm SD) and increases toward the Icelandic Low northwest of the domain (~2.1-cm SD).

Exp_WindVar (Figure 6c) shows that winds drive large sea level variability in the North Sea and around Scotland, Ireland, and Norway (up to 1.4-cm SD). Wind-induced sea level variability is largest in the southern and eastern North Sea (up to ~4.7-cm SD) and is also large in the open ocean. The large SDs in





Figure 6. Sea level variability in different ROMS experiments (Table 1) as the SD (cm) of detrended annual means (1995–2018) for (a) *Exp_AtmosVar*, (b) *Exp_SLPVar*, (c) *Exp_WindVar*, (d) *Exp_BuoyVar*, (e) the linear sum of time series in *Exp_SLPVar*, *Exp_WindVar*, and *Exp_BuoyVar*, and (f) the residual SD of *Exp_AtmosVar* minus the linear sum of components. Isobaths as in Figure 1.

the open ocean and on the shelf are separated by much smaller sea level variability along the shelf break (0.3- to 0.55-cm SD), west of France (0.6-cm SD) and southwest of the United Kingdom (0.5- to 0.7-cm SD).

Exp_BuoyVar (Figure 6d) results in spatially relatively smooth sea level variability that varies from 0.5- to 1.3-cm SD on the shelf. Apart from the intrinsic sea level variability, the magnitude of the variability in the open ocean is comparable to that on the shelf. Its magnitude tends toward 0 at the lateral boundaries because of the repeated annual cycle of GLORYS12v1. The interaction of buoyancy fluxes with river run-off and winds may explain the pattern of buoyancy-driven sea level variability in the German Bight.

Similar to section 4.1, we linearly summed the detrended sea level time series in *Exp_SLPVar*, *Exp_WindVar*, and *Exp_BuoyVar* and computed the resulting SDs (Figure 6e). The residual SDs relative to *Exp_AtmosVar* are less than 0.24 cm on the shelf (Figure 6f). Additionally, the pointwise correlation with the time series of *Exp_AtmosVar* is high (supporting information Figure S3). The residual SD is around 0.23 cm in the German Bight, which indicates that nonlinear feedbacks between buoyancy fluxes and winds, or buoyancy fluxes and river run-off, can have a small effect on interannual sea level variability in this area. The small effect of nonlinear feedbacks suggests that, to first order, the influences of different drivers of sea level variability on the shelf can be studied separately. This justifies the use of multiple (linear) regression methods in observation-based studies (e.g., Calafat & Chambers, 2013; Dangendorf et al., 2013, 2014; Frederikse, Riva, Kleinherenbrink, et al., 2016; Gerkema & Duran-Matute, 2017; Wahl et al., 2013).





Figure 7. Detrended sea level anomalies with respect to time mean sea level (cm) and the ratio of SDs (–) with respect to *Exp_AllVar* (black, dashed) for *Exp_SLPVar* (red), *Exp_WindVar* (green), *Exp_BuoyVar* (purple), and *Exp_OceanVar* (blue) at the coastal locations in Figure 3: (a) Stavanger (NO), (b) Malin Head (IR), (c) North Shields (UK), (d) Cuxhaven (GE), (e) Den Helder (NL), (f) Brest (FR), (g) Santander (SP), and (h) Cascais (PT).

4.3. Relative Magnitudes of Sea Level Variability at Coastal Locations

To investigate sea level variability in each sensitivity experiment, we display the time series for the sample coastal locations of Figure 3 (Figure 7, left panels). Additionally, we calculated the relative magnitudes of interannual sea level variability by dividing the SDs of detrended sea level in each sensitivity experiment by those in *Exp_AllVar* (Figure 7, right panels). These ratios do not necessarily add up to one for a given location, because the sea level variability in one sensitivity experiment may reinforce or interfere with that in another sensitivity experiment.

In most locations, SLP (red) drives the largest interannual sea level variability. This is especially the case for Brest, Santander, and Cascais (Figures 7f–7h, red). In Stavanger, Malin Head, and North Shields, both SLP (red) and wind (green) play a dominant role (Figures 7a–7c). Wind is the primary driver of interannual sea level variability at Cuxhaven and Den Helder (Figures 7d and 7e); there is large overlap between the time series of *Exp_WindVar* (green) and *Exp_AllVar* (black, dashed). The effects of winds at the western and eastern North Sea coasts clearly differ (Figures 7c–7e). Variability due to SLP, buoyancy fluxes, and the lateral boundary conditions is smaller than wind-driven variability at Cuxhaven and Den Helder but not



negligible. Using linear regression, Dangendorf et al. (2013) obtained similar results for the Cuxhaven TG record.

Since the interannual sea level variability in *Exp_OceanVar* is approximately uniform on the shelf (section 4.1), the relative importance of the forcing at the lateral boundaries depends on the pattern of sea level variability in *Exp_AllVar*. Hence, although sea level variability in *Exp_OceanVar* is relatively small (~0.5-cm SD, Figure 5d), the forcing at the lateral ocean boundaries is still a relatively important driver at Stavanger, North Shields, Santander, and Cascais. It results in SD ratios of 0.41, 0.33, 0.36, and 0.42, respectively (Figures 7a, 7c, 7g, and 7h, blue).

Especially Cascais is situated close to the lateral boundaries of our ROMS domain. Because of the nudging layer, sea level variability in *Exp_WindVar* and *Exp_BuoyVar* tends toward the repeated mean annual cycle of GLORYS12v1 at the lateral boundaries (Figures 6c and 6d). Therefore, the SD ratios of sensitivity experiments should be interpreted with caution at locations close to the lateral boundaries. Nevertheless, we expect that the drivers of sea level variability at Cascais, Santander, and Brest are similar, since sea level at these locations is highly correlated in both ROMS and the observations.

4.4. Correlation Between Sea Level at Coastal Locations

To relate the spatially varying effect of different drivers to the correlation between sea level at different locations, we calculated the correlation between detrended annual mean sea level in *Exp_AllVar* at several coastal locations in the region (Figure 8a, lower triangle). In section 4.2, we showed that wind variability is an important driver of interannual sea level variability in the North Sea (Figure 6c). Consequently, sea level along the North Sea coast is highly correlated (e.g., at Wick, Aberdeen, North Shields, Esbjerg, Cuxhaven, Den Helder, and Oostende). Sea level at locations on the southwest of the NWES is also correlated (e.g., Brest, Newlyn, Santander, and Cascais), which is where SLP dominates (Figure 6b). Sea level at Malin Head is strongly influenced by both SLP and winds (Figure 7b) and has high correlations with sea level at Bergen, Stavanger, and the northeast U.K. coastline. Sea level at Malin Head has reduced correlations with sea level at locations in the North Sea, since these are predominantly controlled by wind variability. Correlations between sea level at locations in the North Sea and the English Channel, and in the North Sea and on the southwestern NWES, are low or even negative (e.g., Santander and Brest).

The correlation pattern for *Exp_AllVar* generally agrees with that for PSMSL (Holgate et al., 2013; PSMSL,2020) TGs (Figure 8a, upper triangle), but TG correlations are generally lower, possibly due to data gaps and the influence of local processes. We excluded TGs with less than 18 years of data (75%, gray) and corrected for the nodal cycle following Woodworth (2012), neglecting self-attraction and loading effects. Both ROMS and the TGs show that sea level on the southwestern part of the NWES has low or negative correlations with sea level in the southern and eastern North Sea. This can be explained by the sensitivity of shelf sea level to the NAO (Chen et al., 2014; Tsimplis & Shaw, 2008; Tsimplis et al., 2005; Wakelin et al., 2003; Woolf et al., 2003). For example, a high SLP anomaly at Santander depresses sea level locally. This coincides with stronger southwesterly winds over the North Sea, which induce higher sea level at the North Sea coast through Ekman transport (Tsimplis & Shaw, 2008). This mechanism becomes apparent from *Exp_SLPVar* and *Exp_WindVar* (Figures 6b and 6c).

Complementing Figure 8a, we repeat Figure 4a, now incorporating the effect of SLP (Figure 8b). This leads to increased and statistically significant correlations between sea level at Brest and in the open ocean, because SLP has long correlation scales and the coastal region and open ocean respond to SLP similarly. As discussed earlier, SLP also enhances the contrast between sea level variability on the southwestern part of the NWES and in the North Sea. As a consequence, the coherence of sea level variability along the shelf break extends less far north in Figure 8b than in Figure 4a.

To investigate the role of LSWs inside our domain in driving the coherence of sea level on the NWES, we integrated ERA5 winds along the 400-m isobath from the southern lateral boundary (36°N) to the respective latitudes of Stavanger, Malin Head, Brest, Santander, and Cascais. The LSW integrals are significantly correlated with detrended annual mean sea level in Exp_AllVar (excluding the IB effect) at these locations, which primarily reflect high correlations with sea level in $Exp_WindVar$ (supporting information Table S1). The correlations of sea level with LSWs are higher than those with local winds, supporting the premise of signals propagating along the boundary (Calafat et al., 2012; Sturges & Douglas, 2011). The





Figure 8. Correlation coefficients (-) between (a) detrended annual mean sea level (1995–2018) at several coastal locations across the region for *Exp_AllVar* (lower triangle) and TGs with \geq 75% data available (upper triangle) and between (b) detrended sea level at Brest (o) and at all other grid cells for *Exp_AllVar*. Stipples and isobaths in (b) as in Figure 3.



LSW integral up to Stavanger has relatively low correlations with sea level at North Shields, Cuxhaven, and Den Helder. In line with Figure 8b, including SLP affects the correlation between integrated LSWs and annual mean sea level (supporting information Table S1), reflecting the important role of the IB effect at most locations.

5. Discussion

Our sensitivity experiments showed that atmospheric forcing is the main driver of interannual sea level variability on the NWES. This agrees with previous studies for different time periods (e.g., Dangendorf et al., 2014; Tinker et al., 2020; Wakelin et al., 2003). Additionally, we found that ocean-driven variability is small and coherent on the shelf and combines with atmospheric-driven variability nearly linearly. Similar spatial patterns of sea level variability were obtained by Tinker et al. (2020) with a different regional ocean model, indicating some robustness to model design. Because Tinker et al. (2020) derived boundary conditions from a 200-year control run of HadGEM3 GC3.0 (Williams et al., 2018), the magnitude of sea level variability in their atmospheric and ocean-driven simulations differs from that in *Exp_AtmosVar* and *Exp_OceanVar* (Figures 5c and 5d). Since we used reanalysis data as boundary conditions instead, our results can be compared directly with satellite altimetry but may not be fully representative of (multi)decadal sea level variability.

We have separated the atmospheric forcing into SLP, winds, and buoyancy fluxes for the entire NWES. This complements the analysis of Dangendorf et al. (2014), which focused on TGs in the North Sea. Dangendorf et al. (2014) found that linear regression with winds explains around 50–85% of observed variability from Oostende (Belgium) to Hirtshals (Denmark) and with SLP around 50–65% from Tregde to Bergen (Norway) and Aberdeen to Lerwick (UK). *Exp_WindVar* confirms that wind-driven interannual sea level variability is largest in the southern and eastern North Sea. Our results also show that wind-driven variability is large around Norway, Scotland, and north of Ireland and small on the southwest of the NWES and along the shelf break. In line with Dangendorf et al. (2014), we showed that SLP is the dominant driver of interannual sea level variability along the northeast U.K. coastline. Our sensitivity experiments show that this is also the case on the southwestern part of the NWES. Much of the interannual variability at North Sea TGs is explained with barotropic models (Dangendorf et al., 2014; Piecuch et al., 2019). The baroclinic response to winds could be further investigated by comparing our results to the results of a barotropic setup under the same atmospheric forcing.

Since the NWES seas are shallow, local steric sea level variability is limited (Roberts et al., 2016; Tinker et al., 2020). We showed that buoyancy fluxes result in interannual sea level variability with SDs varying from 0.5–1.3 cm on the shelf (*Exp_BuoyVar*; Figure 6d). At the sample locations in Figure 7, the relative magnitude of buoyancy-driven variability roughly agrees with the finding of Tinker et al. (2020) that the local steric component explains up to 35% of the total variability on the shelf. A large fraction of the steric sea level variability on the NWES in the global model of Roberts et al. (2016) was indeed explained by buoyancy fluxes.

The coherence of sea level along the eastern boundary of the North Atlantic and in the North Sea was previously attributed to wind-driven CTWs (Calafat et al., 2012, 2013; Dangendorf et al., 2014; Sturges & Douglas, 2011). The long correlation scales along the shelf break compared to in the deep ocean (Figures 4c and 4d) suggest that boundary dynamics are indeed at play. This is further supported by the significant correlations of integrated LSWs with simulated coastal sea level (supporting information Table S1), which are similar to the correlations with TG records (Calafat et al., 2012, 2013; Dangendorf et al., 2014). The correlations between integrated LSWs and sea level in *Exp_WindVar* confirm that this results from wind forcing, and the reduced correlations between local winds and sea level suggest that LSWs drive an important part of the variability in *Exp_WindVar*. The correlation between integrated LSWs and sea level in the North Sea is low, mirroring the correlation pattern of a common mode of British TGs with satellite altimetry in (Hogarth et al., 2020) and indicating the dominance of local winds over the North Sea (Figure 6c).

Since wind-driven interannual sea level variability is relatively small at the shelf break and on the southwestern NWES (*Exp_WindVar*, Figure 6c), the coherence of sea level there could partially be explained by the relatively uniform sea level variability due to the forcing at the lateral ocean boundaries and due to buoyancy fluxes (*Exp_OceanVar* and *Exp_BuoyVar*, Figures 5d and 6d). The correlation scales in Figure 4 and the



correlation between sea level and integrated LSWs strongly suggest that wind-driven CTWs influence sea level on the NWES but do not necessarily rule out an effect of open ocean steric anomalies (Chafik et al., 2019), which may be represented in *Exp_OceanVar*. The variability driven by the southern boundary conditions may also reflect northward propagating CTWs triggered south of the domain (Calafat et al., 2012; Fukumori et al., 2015). However, quantifying this remote contribution accurately would require enlarging the model domain and increasing the simulation period.

6. Conclusions

We have used a high-resolution $([1/8]^{\circ} \text{ by } [1/8]^{\circ})$ configuration of ROMS for the NWES to identify the contribution of different drivers of interannual sea level variability during 1995–2018. The interannual sea level variability in our model matches well with that observed with satellite altimetry. Based on sensitivity experiments in which we isolated the effect of atmospheric and oceanic drivers, we conclude that atmospheric variability is the main driver of interannual sea level variability, with the largest SDs in the German Bight (up to 5.6 cm). On the other hand, the forcing at the lateral ocean boundaries causes coherent sea level variability on the shelf, with an SD of only 0.5 cm. The ocean-driven variability is mainly caused by the variation of the western and southern boundary conditions, but its driving physical mechanism requires further investigation. In our model, intrinsic sea level variability has an SD of up to 2.7 cm in the open ocean but is negligible on the shelf.

Our results show that, to first order, the interannual sea level variability induced by variability of SLP, winds, buoyancy fluxes, and the lateral ocean boundaries combines linearly on the NWES. Thus, the contributions of these drivers can be studied separately, which supports the use of multiple linear regression analysis. Wind variability has a large influence on sea level variability in the North Sea and around Scotland, Norway, and north of Ireland. It is the dominant driver in the southern and eastern North Sea, whereas the IB effect dominates at other coastal locations shown in this study. Wind also drives large mesoscale variability in the open ocean, but wind-driven sea level variability is suppressed along the shelf break. At the sample locations in Figure 7, buoyancy-driven sea level variability can have a relative magnitude of up to 55% (North Shields). Sea level variability driven by the lateral ocean boundary conditions can have a relative magnitude of up to 42% (Cascais).

Sea level variability on the southwestern part of the NWES is predominantly driven by the IB effect. As a consequence, detrended annual mean sea level on the southwestern part of the shelf has low or even negative correlations with that in the southern and eastern North Sea, where sea level variability is predominantly driven by local winds. This agrees with the sensitivity of shelf sea level to the NAO (e.g., Tsimplis & Shaw, 2008; Wakelin et al., 2003). The significantly positive correlations between wind-driven sea level variability on the NWES and winds integrated along the continental slope in our model indicate that LSWs and the consequent propagation of signals along the eastern boundary of the North Atlantic are partly responsible for the coherence of sea level variability on the NWES.

To conclude, our results improve the understanding of the drivers of interannual sea level variability on the NWES, which vary in importance depending on location. Explicitly decomposing the contribution of different drivers using ROMS increases confidence in previous observation-based studies and is useful for sea level budget studies. The results of our sensitivity experiments can be used to remove known variability from observations to estimate sea level rise or time of emergence with higher accuracy. This will help coastal decision makers to monitor and detect early warning signals more reliably.

Data Availability Statement

The ROMS input and output files are available at the 4TU.ResearchData repository (http://doi.org/10.4121/uuid:d9656541-ff40-45d0-8859-ac644b155dfb).

References

Amante, C., & Eakins, B. W. (2009). 1 arc-minute global relief model: Procedures, data sources and analysis. NOAA Technical Memorandum NESDIS NGDC-24. National Geophysical Data Center, NOAA. https://doi.org/10.7289/V5C8276M

Acknowledgments

This study has been conducted using E.U. Copernicus Marine Service Information (2019). Neither the European Commission nor ECMWF is responsible for any use that may be made of the Copernicus Information or Data it contains. Dewi Le Bars was funded by EU's Horizon 2020 RECEIPT. Jonathan Tinker was supported by the Met Office Hadley Centre Climate Programme funded by BEIS and Defra. We thank Sybren Drijfhout for his help developing the model. We used Model2roms to derive lateral boundary conditions from GLORYS12v1, and cbrewer and cmocean 2.0 colormaps for our figures.



Andersen, O. B., & Scharroo, R. (2011). Range and geophysical corrections in coastal regions: And implications for mean sea surface determination. In S. Vignudelli, A. Kostianoy, P. Cipollini, J. Benveniste (Eds.), *Coastal altimetry* (pp. 103–145). Berlin, Heidelberg: Springer. https://doi.org/10.1007/978-3-642-12796-0

AVISO (2019). The Ssalto/Duacs altimeter products were produced and distributed by the Copernicus Marine and Environment Monitoring Service (CMEMS). http://www.marine.copernicus.eu

Beckmann, A., & Haidvogel, D. B. (1993). Numerical simulation of flow around a tall isolated seamount. Part 1: Problem formulation and model accuracy. *Journal of Physical Oceanography*, 23, 1736–1753. https://doi.org/10.1175/1520-0485(1993)023<1736: NSOFAA>2.0.CO;2

Calafat, F. M., & Chambers, D. P. (2013). Quantifying recent acceleration in sea level unrelated to internal climate variability. *Geophysical Research Letters*, 40, 3661–3666. https://doi.org/10.1002/grl.50731

Calafat, F. M., Chambers, D. P., & Tsimplis, M. N. (2012). Mechanisms of decadal sea level variability in the eastern North Atlantic and the Mediterranean Sea. Journal of Geophysical Research, 117, C09022. https://doi.org/10.1029/2012JC008285

Calafat, F. M., Chambers, D. P., & Tsimplis, M. N. (2013). Inter-annual to decadal sea-level variability in the coastal zones of the Norwegian and Siberian Seas: The role of atmospheric forcing. *Journal of Geophysical Research: Oceans*, 118, 1287–1301. https://doi.org/10.1002/ jgrc.20106

Chafik, L., Nilsen, J. E. Ø., Dangendorf, S., Reverdin, G., & Frederikse, T. (2019). North Atlantic Ocean circulation and decadal sea level change during the altimetry era. *Scientific Reports*, 9(1), 1041. https://doi.org/10.1038/s41598-018-37603-6

Chapman, D. (1985). Numerical treatment of cross-shelf open boundaries in a barotropic coastal ocean model. *Journal of Physical Oceanography*, *15*, 1060–1075. https://doi.org/10.1175/1520-0485(1985)015<1060:NTOCSO>2.0.CO;2, 8

Chen, X., Dangendorf, S., Narayan, N., O'Driscoll, K., Tsimplis, M. N., Su, J., et al. (2014). On sea level change in the North Sea influenced by the North Atlantic Oscillation: Local and remote steric effects. *Estuarine, Coastal and Shelf Science, 151*, 186–195. https://doi.org/ 10.1016/j.ecss.2014.10.009

Church, J. A., Clark, P. U., Cazenave, A., Gregory, J. M., Jevrejeva, S., Levermann, A., et al. (2013). Sea level change, Climate change 2013: The physical science basis. In T. F. Stocker et al. (Eds.), Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 1137–1216). Cambridge, UK, and New York, NY: Cambridge University Press.

- CNES. (2019). MDT_CNES_CLS18 was produced by CLS and distributed by Aviso+, with support from Cnes. https://www.aviso. altimetry.fr/
- Copernicus Climate Change Service (C3S) (2017). ERA5: Fifth generation of ECMWF atmospheric reanalyses of the global climate. Retrieved October 15, 2019, from https://cds.climate.copernicus.eu/cdsapp#!/home
- Dai, A. (2017). Dai and Trenberth global river flow and continental discharge dataset. Boulder, CO: Research Data Archive at the National Center for Atmospheric Research, Computational and Information Systems Laboratory. Retrieved from https://doi.org/10.5065/ D6V69H1T
- Dangendorf, S., Calafat, F. M., Arns, A., Wahl, T., Haigh, I. D., & Jensen, J. (2014). Mean sea level variability in the North Sea: Processes and implications. Journal of Geophysical Research: Oceans, 119, 6820–6841. https://doi.org/10.1002/2014JC010192
- Dangendorf, S., Mudersbach, C., Wahl, T., & Jensen, J. (2013). Characteristics of intra-, inter-annual and decadal sea-level variability and the role of meteorological forcing: The long record of Cuxhaven. Ocean Dynamics, 63(2–3), 209–224. https://doi.org/10.1007/s10236-013-0598-0
- Dee, D. P., Uppala, S. M., Simmons, A. J., Berrisford, P., Poli, P., Kobayashi, S., et al. (2011). The ERA-interim reanalysis: Configuration and performance of the data assimilation system. *Quarterly Journal of the Royal Meteorological Society*, (April), 553–597. https://doi.org/ 10.1002/qj.828
- Fairall, C. W., Bradley, E. F., Rogers, D. P., Edson, J. B., & Young, G. S. (1996). Bulk parameterization of air-sea fluxes for tropical ocean-global atmosphere coupled-ocean atmosphere response experiment. *Journal of Geophysical Research*, 101(C2), 3747–3764. https:// doi.org/10.1029/95JC03205

Fernandez, E., & Lellouche, J. M. (2018). Product User Manual: For the GLOBAL Ocean Sea Physical Reanalysis Product GLOBAL_REANALYSIS_PHY_001_030.

Flather, R. A. (1976). A tidal model for the northwest European shelf. Memoires de la Societe Royale des Sciences de Liege, 10, 141-164.

Frederikse, T., Riva, R., Kleinherenbrink, M., Wada, Y., van den Broeke, M., & Marzeion, B. (2016). Closing the sea level budget on a regional scale: Trends and variability on the northwestern European continental shelf. *Geophysical Research Letters*, *43*, 10,864–10,872. https://doi.org/10.1002/2016GL070750

Frederikse, T., Riva, R. E. M., Slobbe, D. C., Broerse, T., & Verlaan, M. (2016). Estimating decadal variability in sea level from tide gauge records: An application to the North Sea. *Journal of Geophysical Research: Oceans*, 121, 1529–1545. https://doi.org/10.1002/2015JC011 174. Received, 3

Friocourt, Y., Drijfhout, S., & Blanke, B. (2008). On the dynamics of the slope current system along the West European margin. Part I: Analytical calculations and numerical simulations with steady-state forcing. *Journal of Physical Oceanography*, *38*(12), 2597–2618. https://doi.org/10.1175/2008JPO3744.1

Fukumori, I., Wang, O., Llovel, W., Fenty, I., & Forget, G. (2015). A near-uniform fluctuation of ocean bottom pressure and sea level across the deep ocean basins of the Arctic Ocean and the Nordic seas. Progress in Oceanography, 134, 152–172. https://doi.org/10.1016/j. pocean.2015.01.013

Gerkema, T., & Duran-Matute, M. (2017). Interannual variability of mean sea level and its sensitivity to wind climate in an inter-tidal basin. Earth System Dynamics, 8(4), 1223–1235. https://doi.org/10.5194/esd-8-1223-2017

Greatbatch, J. (1994). A note on the representation of steric sea level in models that conserve volume rather than mass. *Journal of Geophysical Research*, 99(C6), 12,767–12,771. https://doi.org/10.1029/94JC00847

Griffies, S. M., & Greatbatch, R. J. (2012). Physical processes that impact the evolution of global mean sea level in ocean climate models. Ocean Modelling, 51, 37–72. https://doi.org/10.1016/j.ocemod.2012.04.003

- Hermans, T. H. J., Tinker, J., Palmer, M. D., Katsman, C. A., Vermeersen, B. L. A., & Slangen, A. B. A. (2020). Improving sea-level projections on the northwestern European shelf using dynamical downscaling. *Climate Dynamics*, 54(3–4), 1987–2011. https://doi.org/ 10.1007/s00382-019-05104-5
- Hogarth, P., Hughes, C. W., Williams, S. D. P., & Wilson, C. (2020). Progress in oceanography improved and extended tide gauge records for the British Isles leading to more consistent estimates of sea level rise and acceleration since 1958. *Progress in Oceanography*, 184(April), 102333. https://doi.org/10.1016/j.pocean.2020.102333



- Holgate, S. J., Matthews, A., Woodworth, P. L., Rickards, L. J., Tamisiea, E., Bradshaw, E., et al. (2013). New data systems and products at the permanent service for mean sea level. *Journal of Coastal Research*, 29(3), 493–504. https://doi.org/10.2112/JCOASTRES-D-12-00175.1
- Huthnance, J. M. (1981). Waves and currents near the continental shelf edge. Progress in Oceanography, 10(4), 193–226. https://doi.org/ 10.1016/0079-6611(81)90004-5
- Lellouche, J., Greiner, E., Le Galloudec, O., Garric, G., Regnier, C., Drevillon, M., et al. (2018). Recent updates to the Copernicus Marine Service global ocean monitoring and forecasting real-time 1/12 high-resolution system. Ocean Science, 14(5), 1093–1126. https://doi.org/ 10.5194/os-14-1093-2018
- Madec, G., & NEMO team (2016). NEMO ocean engine. https://doi.org/10.5281/zenodo.1472492
- Marchesiello, P., McWilliams, J. C., & Shchepetkin, A. (2001). Open boundary conditions for long-term integration of regional oceanic models. Ocean Modelling, 3, 1–20. https://doi.org/10.1016/S1463-5003(00)00013-5, 1-2
- Miller, L., & Douglas, B. C. (2007). Gyre-scale atmospheric pressure variations and their relation to 19th and 20th century sea level rise. Geophysical Research Letters, 34, 1–5. https://doi.org/10.1029/2007GL030862
- Oppenheimer, M., Glavovic, B. C., Hinkel, J., van de Wal, R., Magnan, A. K., Abd-Elgawad, A., et al. (2019). Sea level rise and implications for low lying islands, coasts and communities. In *IPCC Special Report on the Ocean and Cryosphere in a Changing Climate*.

Orlanski, I. (1976). A simple boundary condition for unbounded hyperbolic flows. Journal of Computational Physics, 21(3), 251–269. https://doi.org/10.1016/0021-9991(76)90023-1

Palmer, M., Howard, T., Tinker, J., Lowe, J., Bricheno, L., Calvert, D., et al. (2018). UKCP18 Marine report. Exeter, UK: Met Office Hadley Centre.

Piecuch, C. G., Calafat, F. M., Dangendorf, S., & Jordà, G. (2019). The ability of barotropic models to simulate historical mean sea level

- changes from coastal tide gauge data. Surveys in geophysics (Vol. 40). Netherlands: Springer. https://doi.org/10.1007/s10712-019-09537-9 Pingree, R. D., & Le Cann, B. (1989). Celtic and Armorican slope and shelf residual currents. *Progress in Oceanography*, 23(4), 303–338. https://doi.org/10.1016/0079-6611(89)90003-7
- Ponte, R. M., & Ray, R. D. (2002). Atmospheric pressure corrections in geodesy and oceanography: A strategy for handling air tides. Geophysical Research Letters, 29(24), 2–5. https://doi.org/10.1029/2002GL016340
- Roberts, C. D., Calvert, D., Dunstone, N., Hermanson, L., Palmer, M. D., & Smith, D. (2016). On the drivers and predictability of seasonalto-interannual variations in regional sea level. *Journal of Climate*, 29(21), 7565–7585. https://doi.org/10.1175/JCLI-D-15-0886.1
- Sérazin, G., Penduff, T., Grégorio, S., Barnier, B., Molines, J.-M., & Terray, L. (2015). Intrinsic variability of sea level from global ocean simulations: Spatiotemporal scales. *Journal of Climate*, 28(10), 4279–4292. https://doi.org/10.1175/JCLI-D-14-00554.1
- Shchepetkin, A. F., & McWillams, J. C. (2005). The regional oceanic modeling system (ROMS): A split-explicit, free-surface, topographyfollowing-coordinate oceanic model. *Ocean Modelling*, 9(4), 347–404. https://doi.org/10.1016/j.ocemod.2004.08.002
- Song, Y., & Haidvogel, D. (1994). A semi-implicit ocean circulation model using a generalized topography-following coordinate system. Journal of Computational Physics, 115(1), 228-244. https://doi.org/10.1006/jcph.1994.1189
- Stammer, D., & Huttemann, S. (2008). Response of regional sea level to atmospheric pressure loading in a climate change scenario. Journal of Climate, 21(10), 2093–2101. https://doi.org/10.1175/2007JCLI1803.1
- Sturges, W., & Douglas, B. C. (2011). Wind effects on estimates of sea level rise. Journal of Geophysical Research, 116, C06008. https://doi.org/10.1029/2010JC006492
- Thompson, K. R. (1986). North Atlantic sea-level and circulation. *Geophysical Journal of the Royal Astronomical Society*, 87(1), 15–32. https://doi.org/10.1111/j.1365-246X.1986.tb04543.x
- Tinker, J., Palmer, M. D., Copsey, D., Howard, T., Lowe, J. A., & Hermans, T. H. J. (2020). Dynamical downscaling of unforced interannual sea-level variability in the North-West European shelf seas. *Climate Dynamics*, 55(7–8), 2207–2236. https://doi.org/10.1007/s00382-020-05378-0
- Tsimplis, M. N., & Shaw, A. G. P. (2008). The forcing of mean sea level variability around Europe. *Global and Planetary Change*, 63(2–3), 196–202. https://doi.org/10.1016/j.gloplacha.2007.08.018
- Tsimplis, M. N., Woolf, D. K., Osborn, T. J., Wakelin, S., Wolf, J., Flather, R., et al. (2005). Towards a vulnerability assessment of the UK and northern European coasts: The role of regional climate variability. *Philosophy*, 363(1831), 1329–1358. https://doi.org/10.1098/ rsta.2005.1571
- Wahl, T., Haigh, I., Woodworth, P. L., Albrecht, F., Dillingh, D., Jensen, J., et al. (2013). Observed mean sea level changes around the North Sea coastline from 1800 to present. Earth Science Reviews, 124, 51–67. https://doi.org/10.1016/j.earscirev.2013.05.003
- Wakelin, S. L., Woodworth, P. L., Flather, R. A., & Williams, J. A. (2003). Sea-level dependence on the NAO over the NW European continental shelf. *Geophysical Research Letters*, 30(7), 1–4. https://doi.org/10.1029/2003GL017041
- Williams, K. D., Copsey, D., Blockley, E. W., Bodas-Salcedo, A., Calvert, D., Comer, R., et al. (2018). The Met Office Global Coupled Model 3.0 and 3.1 (GC3.0 and GC3.1) configurations. *Journal of Advances in Modeling Earth Systems*, 10, 357–380. https://doi.org/10.1002/ 2017MS001115
- Woodworth, P. L. (2012). A note on the nodal tide in sea level records. Journal of Coastal Research, 28(2), 316–323. https://doi.org/10.2112/ JCOASTRES-D-11A-00023.1
- Woodworth, P. L., Pouvreau, N., & Woppelman, G. (2010). The gyre-scale circulation of the North Atlantic and sea level at Brest. Ocean Science, 6, 185–190. https://doi.org/10.5194/os-6-185-2010
- Woodworth, P. L., Teferle, F. N., Bingley, R. M., Shennan, I., & Williams, S. D. P. (2009). Trends in UK mean sea level revisited. *Geophysical Journal International*, 176(1), 19–30. https://doi.org/10.1111/j.1365-246X.2008.03942.x
- Woolf, D. K., Shaw, A. G. P., & Tsimplis, M. N. (2003). The influence of the North Atlantic Oscillation on sea-level variability in the North Atlantic region. *Journal of Atmospheric & Ocean Science*, 9(4), 145–167. https://doi.org/10.1080/10236730310001633803